Author's response to reviews

Title: Individuals with chronic low back pain have greater difficulty in engaging in positive lifestyle behaviours than those without back pain: An assessment of health literacy.

Authors:

Andrew M Briggs (A.Briggs@curtin.edu.au)
Joanne E Jordan (joannejordan14@gmail.com)
Peter B O'Sullivan (P.OSullivan@curtin.edu.au)
Rachelle Buchbinder (Rachelle.Buchbinder@monash.edu.au)
Angus F Burnett (a.burnett@cuhk.edu.hk)
Richard H Osborne (Richard.Osborne@deakin.edu.au)
Leon M Straker (L.Straker@curtin.edu.au)

Version: 4 Date: 14 July 2011

Author's response to reviews:

14th July 2011

The BMC Series Editorial Production Team

Re: BMC Musculoskeletal Disorders, MS: 7587568825110741, Formatting changes required

Dear Sir/Madam

Thank you for your email of 11th July. I have made the formatting changes you requested, as detailed below:

1. I have indicated in the Authors’ Contributions section that all authors read and approved the final version of the manuscript.
2. I have removed shading from Table 1.
3. I have removed tracked changes and highlighting from the paper.
4. I have checked the manuscript for typographical errors.

Thank you for progressing this paper to publication.

Sincerely,

Andrew Briggs